Author's response to reviews

Title: Protocol for systematic reviews of determinants/correlates of obesity related dietary and physical activity behaviours in young children (preschool 0-6yrs): evidence mapping and syntheses

Authors:

Rajalakshmi Lakshman Dr (rl284@mrc-epid.cam.ac.uk)
Veena Paes Mrs (vmp29@medschl.cam.ac.uk)
Kathryn Hesketh Ms (krh40@medschl.cam.ac.uk)
Claire O'Malley (c.i.o'malley@durham.ac.uk)
Helen Moore Mrs (helen.moore@durham.ac.uk)
Ken Ong Dr (Ken.Ong@mrc-epid.cam.ac.uk)
Simon Griffin Dr (Simon.Griffin@mrc-epid.cam.ac.uk)
Esther Van Sluijs Dr (ev234@medschl.cam.ac.uk)
Carolyn Summerbell Professor (carolyn.summerbell@durham.ac.uk)

Version: 4 Date: 16 April 2013

Author's response to reviews:

Please see attached cover letter